Massachusetts Institute of Technology Physics Department Physics 8.334 Spring 03

8.334 questionnaire

Please answer these questions. Your answers will greatly help us to better plan the course, to adjust the pace of lectures, to select appropriate material for the problem sets and term paper topics.

Your Name:

Course/year:

The last 3 physics/math books read:

The last 3 physics courses taken:

Do you use computer for programming? How often? What are your favorite languages?

What is your field? What part of Stat Mech interests you most? Do you see any relevance for your research?